



Response to reviewers

Emerging Markets Case Studies: Name of case study

Summary comments to the reviewers and editors, typically expressing appreciation for the quality of the review, and commenting on the extent to which the recommended revisions have been made.

Reviewer 1 feedback	Authors' response
Copy each point made by Reviewer 1 in a separate row. Create as many rows as are required for the feedback points.	Detail exactly how the authors have addressed each point made in the left-hand column. Include details of how exactly the case or TN may have been revised, and mention the page in the revised submission where this change was made. If no changes were made based on the reviewer or editor's feedback, explain fully why the authors chose not to make the recommended changes.

Reviewer 2 feedback	Authors' response

Reviewer 3 or Editor feedback	Authors' response